Author’s response to reviews

Title: A searchable database of medical education objectives - creating a comparable gold standard

Authors:

Sage Arbor (sagearbor@gmail.com)
Sam Brooks (sbrks2013@gmail.com)
Namita Biala (namita.biala@gmail.com)

Version: 3 Date: 25 Jan 2018

Author’s response to reviews:

The objectives I used in the database for this article were either attained from publications under a creative commons license or I directly attained approval from the societies which had published objectives. I don't see the declaration for Consent to Publish section but will try to add this verbiage there as well. Im hoping it appears after I hit next.